**Author’s response to reviews**

**Title:** A Randomized, Controlled Clinical Trial: The effects of Mindfulness-Based Cognitive Therapy on Generalized Anxiety Disorder among Chinese community patients: protocol for a randomized trial

**Authors:**

- Samuel YS Wong Prof. (yeungshanwong@cuhk.edu.hk)
- Winnie WS Mak Prof. (wwsmak@psy.cuhk.edu.hk)
- Eliza Cheung Ms (eliza.cheung@cuhk.edu.hk)
- Candy Ling Ms (candy.ymling@gmail.com)
- Wai Sze Lui Ms (wacylui@yahoo.com)
- Wai Kwong Tang Prof. (tangwk@cuhk.edu.hk)
- Rebecca Wong Ms (rebeccawong@cuhk.edu.hk)
- Herman Lo Mr (hkfws_hlo@yahoo.com.hk)
- Stewart Mercer Prof. (stewart.mercer@glasgow.ac.uk)
- Shuk-wak Helen Ma Ms (shelenma@hkucc.hku.hk)

**Version:** 3  **Date:** 9 November 2011

**Author’s response to reviews:**

Dear Editor,

Ref: 2109417371481657

A randomized, controlled clinical trial: The effects of mindfulness-based cognitive therapy on generalized anxiety disorder among Chinese community patients: protocol for a randomized trial

We are pleased to submit the revised manuscript.

Thank you very much for your kind attention.

Sincerely,

Samuel YS Wong.
Professor, The Chinese University of Hong Kong.